CENWP-OP-B                                                                                                                                     15 June 2004

MEMORANDUM FOR CENWP-OP, ATTN: Gary Johnson

SUBJECT: Bonneville Lock and Dam, Fishway and Fish Activities for week 24 of 2004, which covers the period 6 June through 12 June 2004.

1.  OPERATION SUMMARY.

1.1 Daily river flow averaged 259.1 to 285.8 kcfs.  Daily project tailwater elevation averaged 19.4’ to 20.8’ msl.  Daily project forebay elevation averaged 74.8’ to 76.0’ msl.  Daily river temperature averaged 57 to 58° F.  Daily turbidity measured 2.5 to 5.0 on the secchi disk.  

1.2 Spill Bays 1 and 18, which supply attraction flow for Cascades Island and B Branch respectively, are part of the FPP spill pattern.  Spill averaged 94.2 to 99.3 kcfs.    
1.3 Unit operation.  

1.3.1 Unit 1 remains out of service for turbine rehab.

1.3.2 Units 7-10 were out of service from 0630 until 1730 daily during 7 June through 11 June to facilitate PH 1 headgate delivery.    

1.3.3 Unit 15 went to manual 3D cam adjustment at 0933 on 5 April.  New modules are on order.   

1.4 Second Powerhouse fish units provide attraction flow for the Washington Shore fish ladder.  The units are placed on standby from approximately 0000-0300 nightly if drawdown exceeds 1.0’.  

1.5 Gatewell drawdowns are measured on Wednesdays.  Drawdowns are measured daily on Unit 17.

2.  ACTIVITIES. 
2.1 AWS Closures.  No AWS valves were closed this reporting week.

2.2 STS/VBS Inspections.  Unit 17 B and C VBS’s were cleaned on 09 June. 

2.3 Dewatering and Fish Salvage.  No dewatering or salvage activities occurred this reporting week. 

3.  RESEARCH.
3.1 Battelle/BAE Systems FGE testing.  Data collection continues.     

3.2 CRITFC.  Sampling twice a week.

3.3 ERDC TDG.  Sensors remain installed in the tailrace.

3.4 ODFW Pikeminnow Tagging.  Electrofishing in Bonneville’s forebay and tailrace will continue on 28 June through 2 July.

3.5 NMFS.  No fish were collected for NMFS researchers during the week.

3.6 NOAA FISH lamprey.  Lamprey ladder installation began 24 May.  

3.7 Pelfrey’s Sturgeon Hatchery.  Capturing breeding size sturgeon in Bonneville’s tailrace daily from 5/28 – 6/28.

3.8 PSMFC SMF.  Seven steelhead and five smolt were observed sliding over the separator bars this week.     

3.9 USGS FPE testing.  Last fish release occurred on 3 June for the spring study.  Summer study to begin 20 June.  

3.10 University of Idaho.  Sampling daily in the AFF.

3.11 Wildlife Services/Avian Abatement.  Birds were hazed this reporting week with rockets, cracker shells, and 15mm pyrotechnics.  None were lethally taken.

3.12 WDFW Fish Counting.  16-hr visual counts at Washington Shore and Bradford Island.

4.  FISHWAYS 
4.1 Project biologists inspected 6 June through 12 June this reporting week.      

4.2 Fish Passage Plan Observations.

	DATE
	LOCATion
	FPP Deviation
	CAUSE
	RESPONSE
	TR#

	All week
	PH1 PCC
	Velocity <1.5 fps
	Meter in slowest part of the channel
	This is a normal condition.
	

	All week
	FG3-4
	Incorrect position for TW
	Bent shaft
	No action taken.
	

	All week
	PH2 PCC
	Velocity < 1.5 fps
	No acoustic sensor
	Will repair during winter 04/05
	

	6/7 and 6/10 
	BI weir
	Weir reading 1.1 
	Unknown
	Operators to adjust 
	

	6/8
	Spillbay 18
	Gate position > 2.0
	Air valve shut.  
	Electricians called

See Misc 6.1
	

	6/7-9 and 6/11-12
	Weir 37
	Weir reading >1.3’ 
	Shad 
	Operators to adjust valve See Misc 6.2
	

	6/10
	B Branch 
	>2.0 Differential
	FV 4-3 not controlling properly
	TR submitted and FV 4-3 in manual  See 4.6.1
	04-21750

	All week
	Cascades Island 
	In manual
	Computer software issue
	TR submitted 

See  4.6.2
	04-21382

	
	
	
	
	
	


4.3 Fish Ladder Temperatures.   Measurements are taken three weirs down from the count station at Bradford Island and at the junction of the UMT and the main ladder at Washington shore.  Temperatures are in Fahrenheit.

	Date
	Wash. Shore
	Bradford Is.

	6/6
	60.2
	60.4

	6/7
	60.0
	60.1

	6/8
	60.1
	60.2

	6/9
	60.3
	60.4

	6/10
	60.5
	60.6

	6/11
	60.1
	60.1

	6/12
	60.0
	60.1


4.4 Sensor Calibration.  Calibration was checked on 9 June this reporting week.

4.5  Zebra Mussel Monitoring.  No signs of colonization were observed this reporting week. 

4.6 Fishway Status.  

4.6.1 The B Branch ladder elevation sensor was giving an incorrect reading that controls FV 4-3 gate opening.  FV4-3 is operating in manual.  TR has been submitted to the electricians.  

4.6.2 Cascade Island has not been properly controlling in Auto all week due to a software issue.    TR has been submitted and electricians are troubleshooting the system.     

5 CONSTRUCTION

5.1 PH 2 Corner Collector Hi-Q PIT tag detector- Prototype antenna construction started on Cascades Island.  If feasible, the prototype antenna will then be constructed into the existing channel during the in-water work period during FY05.  See Misc. 6.3 

5.2 SMF hydrocannons.  The hydro cannons at the SMF are not operational at this time.  The Project is working on repairs.    See Misc 6.4
5.3 Adult PIT tag exit detectors.  Current schedule is set for installation in WA ladder for Winter 04/05.

5.4 Head Gate Delivery. Delivery occurred from 7 June through 11 June at Powerhouse 1.    Task completed 6 days ahead of schedule.    

5.5 PH 1 Gantry Crane Delivery and Erection.  Scheduled for 9 August through 4 December.    

6 MISCELLANEOUS AND OTHER NOTEWORTHY ITEMS

6.1 Spillbay 18 was inoperable from 0420 until 0742 on 8 June.  Electricians discovered the air valve to the gate dogs was accidentally closed.  During that time, spillbay 17 was used to compensate and maintain 75 kcfs total spill.     

6.2 Shad appear to be clogging the bleed off valve for weir 37.  Operators to blow air through the screen twice daily and structural crew to rake on Monday’s and Thursday’s if needed. 

6.3  PIT antenna scheduled for delivery at PH 2 forebay on 24 June.  

6.4 Repairs to the avian cannon are scheduled for completion by the end of June.  Wildlife services will conduct full time hazing activities until 26 June.   

6.5 The Project had released approximately 10 gallons of turbine oil into the river by the old navigation lock on 5 June.  Once the final report is available, I will send out an official memorandum with the details.  

6.6 On 12 June, a new Teletype indicating an increase to 120% TDG for the evening to 175 Kcfs.  

6.7 Oil sheen was reported in 8C gate slot on 12 June.  Approximately 10% of the slot was covered and absorbent pads were deployed to soak up the oil.   

JAMES R. MAHAR, P.E.

Operations Manager
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